

GOPEN ACCESS

Citation: Clark J, Davis EL, Prada JM, Gass K, Krentel A, Hollingsworth TD (2023) How correlations between treatment access and surveillance inclusion impact neglected tropical disease monitoring and evaluation—A simulated study. PLoS Negl Trop Dis 17(9): e0011582. https://doi.org/10.1371/journal.pntd.0011582

Editor: Victoria J. Brookes, The University of Sydney School of Veterinary Science, AUSTRALIA

Received: September 8, 2022

Accepted: August 9, 2023

Published: September 6, 2023

Copyright: © 2023 Clark et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its <u>Supporting</u> <u>Information</u> files. All code is available at <u>https://</u> github.com/iamjessclark/non_access.git.

Funding: This work was supported by funding awarded to the Neglected Tropical Disease Modelling Consortium (DTH) by the Bill & Melinda Gates Foundation [OPP1184344]. The funders had no role in study design, data collection and RESEARCH ARTICLE

How correlations between treatment access and surveillance inclusion impact neglected tropical disease monitoring and evaluation— A simulated study

Jessica Clark^{1,2}*, Emma L. Davis², Joaquin M. Prada³, Katherine Gass⁴, Alison Krentel^{5,6}, T. Déirdre Hollingsworth²

 School of Biodiversity, One Health & Veterinary Medicine, University of Glasgow, Glasgow, Scotland, 2 Big Data Institute, Neglected Tropical Disease Modelling Consortium, University of Oxford, Oxford, England,
School of Veterinary Medicine, Faculty of Health and Medical Sciences, University of Surrey, Guildford, England, 4 Neglected Tropical Diseases Support Center, Task Force for Global Health, Decatur, Georgia, United States of America, 5 School of Epidemiology and Public Health, University of Ottawa, Ottawa, Canada, 6 Bruyère Research Institute, Ottawa, Canada

* Jessica.clark@glasgow.ac.uk

Abstract

Neglected tropical diseases (NTDs) largely impact marginalised communities living in tropical and subtropical regions. Mass drug administration is the leading intervention method for five NTDs; however, it is known that there is lack of access to treatment for some populations and demographic groups. It is also likely that those individuals without access to treatment are excluded from surveillance. It is important to consider the impacts of this on the overall success, and monitoring and evaluation (M&E) of intervention programmes. We use a detailed individual-based model of the infection dynamics of lymphatic filariasis to investigate the impact of excluded, untreated, and therefore unobserved groups on the true versus observed infection dynamics and subsequent intervention success. We simulate surveillance in four groups-the whole population eligible to receive treatment, the whole eligible population with access to treatment, the TAS focus of six- and seven-year-olds, and finally in >20-year-olds. We show that the surveillance group under observation has a significant impact on perceived dynamics. Exclusion to treatment and surveillance negatively impacts the probability of reaching public health goals, though in populations that do reach these goals there are no signals to indicate excluded groups. Increasingly restricted surveillance groups over-estimate the efficacy of MDA. The presence of non-treated groups cannot be inferred when surveillance is only occurring in the group receiving treatment.

Author summary

Mass drug administration (MDA) is the cornerstone of control for many neglected tropical diseases. As we move towards increasingly ambitious public heath targets, it is critical to investigate ways in which MDA weaknesses can be strengthened. It is known that some analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

individuals systematically choose not to participate in treatment. It is also becoming evident that others systematically do not have access to treatment. What is less clear however, is how access to treatment correlates to inclusion in surveillance efforts, and in turn, how this impacts the monitoring and evaluation of intervention programmes. If individuals with access to treatment are more likely to be included in surveillance efforts, then this implies that those without access to treatment are likewise more likely to be excluded from surveillance. Extending the individual-based lymphatic filariasis model, TRANSFIL, we show that exclusion to treatment and surveillance negatively impacts the probability of reaching public health goals, though in populations that do reach these goals there are no signals to indicate excluded groups. Increasingly restricted surveillance groups over-estimate the efficacy of MDA. The presence of non-treated groups cannot be inferred when surveillance is only occurring in the group receiving treatment.

Introduction

Neglected tropical diseases (NTDs) are a heterogeneous group of diseases listed by the World Health Organization (WHO) as causes of great human, social and economic burden, largely impacting vulnerable marginalised communities in tropical and subtropical regions [1]. Reducing the burden of infection for some of these diseases has been achieved through mass drug administration (MDA) which refers to the distribution of medicine to an entire eligible population or specific subset (for example, those over a certain age) within a given administrative area regardless of individual infection status [1]. It is widely accepted that the treatments chosen for MDA have a good safety profile, such that there is little risk from treating uninfected people. In addition, test and treat methods can be resource costly, where MDA can reduce costs significantly. This is particularly relevant for high prevalence diseases/ locations and for diseases where diagnosis is more difficult or invasive (for example onchocerciasis where skin snip biopsies must be conducted, often in multiples to increase diagnostic sensitivity [2]). Based on estimates of prevalence before and/or during a programme within administrative units, the WHO provides recommended treatment frequencies and coverage levels. These have recently been updated for the next decade in the 2021–2030 roadmap [1]. Unfortunately, for the NTDs where MDA is the cornerstone of control, achieving recommended coverage levels remains an ongoing challenge that negatively impacts progress towards elimination as a public health problem (EPHP) [1,3].

Sub-optimal treatment coverage can be due to a lack of access to treatment opportunities, or a lack of adherence to treatment regimens. Recent efforts have been made to unify the terminology used by the global community, with 'never treatment'covering both routes of missed or never received treatment [3]. For the purposes of this analysis, we need to delineate the two so that we can be clear on which types of behaviour we are representing in the model. A lack of treatment access (exclusion) can occur for many reasons, including impractical timing of treatment delivery (for example during the day for those away at work), because community treatment distributors do not go to certain homes or areas, or because people themselves cannot access facilities where treatments are distributed from [4,5]. We refer to this as systematic non-access and the group without access as being excluded. It is also possible that inclusion in surveillance efforts is correlated to treatment access, such as with rabies [6], schistosomiasis [4] and lymphatic filariasis [7]. In this instance, people with poor access may be excluded from both treatment and surveillance efforts—influencing estimates of prevalence, and subsequent programme decision making. For several NTDs, modelling has shown that sufficient treatment coverage should suppress transmission to a breakpoint [8,9] below which transmission is no longer sustainable [10]. However, if a sufficiently large sub-population are not receiving treatment and are not accounted for in surveillance efforts but are nonetheless contributing to transmission, target breakpoints may not be reached, or may be perceived as reached but not maintained-driven by this undetected reservoir of infection. It is therefore critical to understand the impact of such groups and whether the level of exclusion can be inferred.

The other commonly cited reason for lower than target coverage, is non-adherence. In this work, this refers to the successful delivery and distribution of treatment to diverse populations, but an active decision by an individual to not participate/ ingest medication or use it as intended [11–14]. This is in contrast to matters of treatment access, which is not a choice made by the recipient. Where this non-adherence choice is made repeatedly over successive rounds of MDA, we term this systematic non-adherence. Reasons for this behaviour are numerous, including (but not limited to) fear of side effects [15,16], belief systems [17] and misinformation regarding safety (for example, schistosomiasis treatment with praziquantel during pregnancy) [18,19]. Quantifying the proportion of the population this pertains to is somewhat possible through behavioural questionnaires giving better insight into who is actually being treated [15,20]. Evidence suggests that treatment registers are often incomplete highlighting the need for other methods to detect 'never treated' individuals [21], which is ultimately vital to accurately estimating the efficacy of intervention programmes based largely on MDA.

Of the five diseases that rely on MDA for control, the WHO identified improving access and delivery as a critical action point for lymphatic filariasis. As a case study, we extend the highly detailed individual-based model, TRANSFIL [22,23], simulating the infection dynamics of lymphatic filariasis at the level of the individual human host. We use this model to investigate the impact of unmeasured, untreated, and therefore unobservable groups, on "true" versus observed (i.e., measurable by surveillance) infection prevalence in humans. Using simulated populations with a range of baseline prevalences, we assessed the probability of reaching EPHP targets as a function of exclusion proportions. We also provide estimates of the number of rounds of treatment needed to reach these targets in the presence of exclusion. Finally, we investigated whether it was possible to relate surveillance insight to the exclusion proportion. We simulated the transmission assessment survey (TAS), conducted after at least five years of effective treatment [24], which is used by lymphatic filariasis programmes to determine whether MDA can be stopped. As the first evaluation survey required of lymphatic filariasis programmes, this is the time point at which challenges in achieving the goals are most frequently detected [25].

Methods

The TRANSFIL model framework has been described in detail and validated in various populations and settings elsewhere [23]. Parameter values are given in supplementary material, where a link to all code can also be found. In short, in this stochastic model, the male and female worm burdens within each human host are modelled, with fertile female worms producing microfilaria (mf), and with the density of infectious third stage larvae (L3) in the human-biting mosquito population reflecting the overall microfilaraemia density in the human population and the infection pressure. Demographic characteristics of each human host are modelled including age, sex and exposure risk. Here, we present simulations relevant to most of Africa where we consider *Anopheles*-driven transmission and treatment with Ivermectin and Albendazole. We do not simulate *Culex*-driven transmission that simulates Indialike populations where treatment is with Diethylcarbamazine and Albendazole treatment, however, we expect qualitatively similar results with respect to exclusion.

Simulated populations

We simulated 35,000 populations with baseline prevalences from 0 to ~60% (Fig A in S1 Supplementary Material). Prevalence was modulated via the vector-to-host ratio and the bite rate aggregation parameters (k) with increasing prevalence associated with higher vector-to-host ratios and more randomly distributed bites (Fig B in S1 Supplementary Material; parameters in Table A in S1 Supplementary Material). Populations were filtered to have a minimum of baseline 1% prevalence across the population over the age of five. There were a total of 14,145 populations used in the simulations. The same populations were reset to their baseline state and used in each of the diagnostic/ access scenarios.

Simulated diagnostics

To determine the impact of non-access on prevalence estimates and treatment efficacy we observed prevalence in four groups, using the two main diagnostic methods available. When using the filariasis test trip (FTS) diagnostic, the operational target prevalence that indicates elimination as a public health problem is 2%. When using detectable mf prevalence from bloody smears (or similar), the EPHP indicated prevalence is 1%

FTS diagnostic: We simulated FTS diagnostic use in three groups with target EPHP prevalence as above; 1) the whole population > five-years-old (referred to as True Prevalence), 2) sixand seven-year-olds (referred to as TAS, see ref [24]), 3) in those aged > five-years-old and with access (Community-wide with access). We assumed male or female worm presence would be detected with a 93.1% sensitivity [26] without considering false positives from coinfection.

Blood smear (or similar): Based on expert advice from co-author Dr Katherine Gass, as has been discussed in decision-making settings, we simulated detectable mf prevalence in those >20-year-olds (referred to as mf TAS).

Simulated treatment

To investigate the probability that five years of treatment would result in achieving the desired threshold, after 100 years burn-in populations received treatment for five years, with an additional treatment at the sixth year, as is common due to the timing of the pre-TAS [24]. We explored the dynamics under no exclusion (where exclusion refers to treatment access and inclusion in surveillance), 10% exclusion, and 50% exclusion (model methods below). Whilst 50% exclusion may seem high, evidence from the lymphatic filariasis literature suggests that treatment repeatedly failed to reach certain homes, resulting in a large percentage of treatment non-access [11,15,17,27]. This problem is not confined to lymphatic filariasis (e.g., schistosomiasis [4,5]), and given the translatability of these ideas to other models, this justifies the investigation into the impacts of such a high exclusion percentage.

To investigate the dynamics once treatment stopped when the appropriate thresholds were met, we recorded prevalence in our observed groups every month for 12 years. We present this part of our analysis focussing on the populations that reached the threshold targets by the fifth MDA. To investigate the number of rounds of treatment needed to reach the operational targets as a function of baseline prevalence and the exclusion proportion in our observed populations, treatment was continued until the target was reached or the simulation was allowed to continue for 40 years at which point the simulation would stop and note the failure to reach the target.

Surveillance inclusion, treatment access, & adherence

The model framework was adapted to investigate the impact of a correlation between treatment access, and inclusion in surveillance. All populations were targeted with 65% treatment coverage that was applied to the observable surveillance groups with access. Treatment access and inclusion in surveillance had a far stronger impact on infection dynamics than adherence (supplementary material Fig C in S1 Supplementary Material). We therefore focus on the impact of increasing exclusion in the results. These and all subsequent investigations used a moderate level of non-adherence (0.2) in line with previous investigations [28].

We partitioned all populations into two groups; access and non-access. Those in the nonaccess group did not receive treatment and were not included in relevant prevalence estimates. The access group were the target of the 65% coverage. This allocation occurs when each individual is born, and is generated from a binomial distribution parameterised for a Bernoulli draw (i.e., N trials = 1). This allocation is static for the duration of the simulation. This partitioning of the population is akin to adherence in previous work on lymphatic filariasis [29] onchocerciasis [30], human African trypanosomiasis [31] or hookworm [32]. Additionally, we assume homogenous mixing between individuals and vectors, regardless of their treatment access, such that those with no access still contribute to transmission (Fig 1) and therefore the overall infection dynamics of a community. The implications of this assumption are addressed in the discussion.



Fig 1. Model schematic. A. Adapted from Dyson et al [14] visualising how varying degrees of adherence to treatment from 0 to 1 impact treatment participation over successive rounds. The background colour represents the probability that a person will participate in MDA from white (no participation) to dark blue (definitely participating). We augment these with population partitioning into those who have access to treatment (those in white or blue) and those with no access to treatment (grey). B. The proportion of the population that have received 1, 2, 3, 4, or 5 rounds of treatment after five years as a function of example values of adherence (0, 0.2 and 1) and non-access (0%, 10% and 50%) experiencing 85% target coverage to the with-access population. Human outlines attributed to Microsoft clipart.

https://doi.org/10.1371/journal.pntd.0011582.g001

To account for treatment adherence, each individual >five-years-old in the treatment access group was assigned a probability of treatment. The group excluded from treatment and surveillance have 0 (zero) probability of being treated. For those with access to treatment, the probability of treatment is a function of the overall target treatment coverage and the correlation in adherence to treatment across rounds, as derived by Dyson et al [14]. In short, as shown in Fig 1A, when there is a weak correlation in treatment participation between rounds (systematically low adherence), treatment participation across the population is more randomly distributed, such that eventually, over successive rounds everyone is likely to be treated at some point. Alternatively, when the correlation between who gets treated from one round to the next is high, adherence becomes systematic, such that whether someone participates in treatment is determined by their behaviour in the first round-if they did not participate in round one, they will not participate in any subsequent rounds. The more systematic treatment adherence is, the more individuals there are that participate in all five rounds of treatment, but also the more individuals there are that are rarely-if ever-treated (Fig 1B). However, as more people are excluded from treatment with declining access the proportion of the total population who receive any treatment at all declines (Fig 1B), such that the effective overall coverage (% of the population) is lower. This is most notable when adherence is systematic.

With the adaptations to this model, we investigated the probability that the normal routine of MDA would result in control (reaching EPHP) given the population observed and extent of non-access. We then investigated how many rounds would be necessary to reach EPHP. Finally, to observe infection dynamics when treatment programmes cease, we next selected populations that had reached the EPHP indicator for each surveillance group, by year five when surveillance most often occurs.

Results

Impact of exclusion on reaching EPHP targets

We observed the probability of achieving elimination as a public health problem after six years of treatment, as a function of the baseline true prevalence and the proportion excluded from treatment and surveillance (Fig 2). The number of populations in each baseline prevalence bin are shown in Fig 2. A total of 14,145 of the simulated populations were used, however we capped Fig 2 observations at 34% prevalence as per data from previous work [23].

There are three main observations; first, exclusion reduces the probability of reaching EPHP even in very low prevalence locations (Fig 2 True Prevalence). With a starting prevalence of 1–5%, five years of MDA yields just over a 50% probability of achieving EPHP when treatment is provided as intended to a whole community, with everyone who is treated also participating in surveillance–essentially whether 65% coverage will achieve EPHP is as unlikely as it is likely. This reduces significantly with increasing levels of non-access (Fig 2, True Prevalence red and blue versus orange).

Secondly, the perceived probability of achieving EPHP is driven in part by the population in which you look. This is most pronounced in lower prevalence communities as infections are easier to miss. Increasingly narrowed observed groups overestimate the probability of reaching targets (comparing between panels Fig 2). For example, community-wide surveillance in those with access to treatment, even in the presence of exclusion captures that five years of treatment will not reach EPHP when prevalence is >5%, whilst observations in only those >20-years-old suggests ~ 10% chance in the same baseline prevalence populations. This difference is exaggerated in the TAS observation, where EPHP appears possible to achieve even in populations where prevalence is >10%.

This suggests, as our final observation, that regardless of the level of exclusion, TAS does not reflect the dynamics of the wider population (Fig 2, True Prevalence vs TAS panels),



Fig 2. The probability of achieving elimination as a public health problem after five years of treatment as a function of exclusion and the baseline prevalence. The number of populations in each baseline prevalence group is shown on the x-axis. The x axis is the true prevalence in the whole population >five-years-old. The target for all panels apart from the mf TAS is $\leq 1\%$ prevalence, whilst the target for mf TAS was $\leq 2\%$. Exclusion is shown in orange for 0% exclusion, red for 10% exclusion, and blue for 50% exclusion. We assume a moderate correlation value for adherence to treatment of 0.2 [14].

https://doi.org/10.1371/journal.pntd.0011582.g002

reaching EPHP indicator targets despite not actually having done so in the wider population. There is a significantly higher probability of appearing to have achieved the EPHP threshold using TAS with FTS in six- and seven-year-olds. This is particularly evident the lower the prevalence. With no exclusion, there appears to be a 25% probability that the target will be (perceived to be) reached in populations with a true starting prevalence of 10–14%. However, in all other observation groups, regardless of exclusions to treatment, by 10–14% prevalence, the baseline prevalence is too high to control infection in five years even with full access to treatment.

We next investigated the number of rounds of treatment needed to reach the appropriate elimination targets for each surveillance group as a function of exclusion. It was hoped that by cross referencing the true baseline prevalence with the number of years of treatment necessary to reach the EPHP target, and exclusion, this could provide an indicator of the true baseline prevalence and the level of exclusion present in a population when treatment had been ongoing for more years than was projected to be necessary. However, if surveillance is occurring only in the group being treated it is not possible to detect the presence of excluded groups (Fig 3). The group being observed continues to have a strong impact on the dynamics observed. For example, in Fig 3 the >20-year-old group estimates that populations with prevalence up to 10% could be largely controlled within five years, whilst in the true prevalence observations even 10% prevalence would require a minimum of five years up to 15–20 years to control with MDA. Similarly, the mean behaviour of the TAS routine suggests control would be possible even at the highest prevalences on average within 10–15 years, whilst the true observations suggest this is more like 25 years. The higher the prevalence the more uncertain the number of years necessary to control infection.

Impact of exclusion on maintaining EPHP

Finally, we observed the infection dynamics across the four observed surveillance groups as a function of the excluded proportion of the population. Based on FTS, infection prevalence was observed as: 1) true prevalence in the whole population aged \geq five-years-old 2) measured prevalence in everyone over five-years-old with access to treatment and therefore included in surveillance 3) as observed during the TAS, in six- and-seven-year-olds with access to treatment. Prevalence quantified from detectable mf was used in group 4) in over 20-year-olds with access to treatment and included in surveillance. This is shown for 0%, 10%, and 50% exclusion. The number of populations used for each simulation (i.e., that were used to generate each line of Fig 4) is shown in Table B in S1 Supplementary Material.

Here, we make two observations; first, the extent of exclusion has little impact on prevalence in those populations that do reach EPHP, and the maintenance of EPHP once treatment stops (Fig 4), though the prevalence is indeed lower in populations that have 50% exclusion and still successfully reach EPHP. To less of a degree this is also the case for those with 10% exclusion (Fig 4 True Prevalence panel).

Secondly, there is little difference between exclusion groups in the overall mean behaviour of the dynamics, however there is a distinct increase in variation around the mean dynamics when observing infection with standard TAS methods, which is most pronounced the 50% exclusion group after treatment stops. Again, this suggests that often, the TAS method will not reflect true population dynamics.

Discussion

In this modelling investigation we illustrated how the correlation between treatment access and surveillance exclusion can impact the success of, and the M&E for, intervention



Fig 3. The number of treatment rounds needed to reach the elimination as a public health problem target (1% for the mf TAS and 2% for everything else). This has been capped at 20 years of treatment.

https://doi.org/10.1371/journal.pntd.0011582.g003



Fig 4. Prevalence over 12 years with standard deviation of the mean observed in four surveillance groups as a function of treatment and surveillance exclusion. True prevalence refers to prevalence in the whole population \geq five-years-old. Community-wide with access is also the population \geq five-years-old but only in those with access to treatment and surveillance. TAS simulates the observations made in six- and-seven-year-olds with access to treatment. Finally, the mf TAS quantifies prevalence by detectable mf in >20-year-olds. Exclusion is represented in orange for 0%, red with green standard deviation for 10% and blue for 50%. As described

in the methods, treatment is given for five years plus one likely given due to the timing and the way treatment is ordered and distributed. The populations used in these simulations had reached elimination as a public health problem after five years of treatment. Note the different limits on the y axes.

https://doi.org/10.1371/journal.pntd.0011582.g004

programmes. A repeated lack of access to treatment and exclusion from surveillance impedes the probability that a population will reach the EPHP target after five years of treatment, but it does not impact the maintenance of it in those populations that do reach the target and subsequently stop MDA. Five years of treatment is not sufficient to control the disease in most cases; even when treatment is available to 65% of the eligible population and surveillance captures all the infections in the treated population, there is only a 50:50 chance the MDA programme will succeed. In addition, the TAS is not an appropriate tool for quantifying population-level dynamics. Treatment of, and observations in only six-and-seven-yearolds does not control infection and overestimates the probability of MDA success, with significantly more variation around the mean dynamics in the results obtained from the TAS, which increases with increasing exclusion. Across all these observations it is also evident that the observation group has a strong impact on the estimated dynamics. Intuitively, the more constrained the group under observation, the less the observed dynamics reflect that of the wider population. Crucially, we show that excluded groups cannot be detected when observations are made only in the surveillance group with access to treatment. As MDA is the cornerstone of control for five of the WHO NTDs, all with ambitious prevalence targets set for 2030, our results highlight the need for accurate coverage estimates and the importance of accessing demographic groups that may be easily overlooked in treatment and surveillance efforts.

When programmes are evaluating their progress, the discrepancies sometimes seen between coverage survey numbers and treatment numbers can indicate when those conducting surveys and those distributing treatment have access to different individuals, groups, or communities [33], suggesting groups with non-access. This is driving a larger discussion about how to identify, measure and engage with this group of untreated individuals [3,34]. Our analysis highlights the importance of designing surveys that aim to minimise bias. By identifying the people that make up these discrepancies, we are potentially also more likely to increase the uptake of MDA treatments. For example, it has been noted that there is a distinct intersection between NTD treatment access and disabilities, particularly when NTDs like filariasis or leprosy cause stigmatising morbidity and disabilities, impacting treatment reach [7]. There is also evidence that access to treatment for one disease is correlative to healthcare access more broadly, including other treatment programmes like malaria [35], meaning if a person has no access to treatment for one disease, it is likely they do not have access to many treatments. Additionally, whilst evidence suggests that women are less adherent to treatment, men are often outside of the home at the time of MDA and at the time of CES and surveillance [21].

There are a number of limitations and simplifying assumptions in this analysis. In the absence of non-access, the model assigns a bite risk to each individual, drawn from a population-level gamma distribution (Fig 5A). As a function of the way in which the population is separated into those with access to treatment and those without, we maintain the assumption that everyone, regardless of treatment access, has a bite risk drawn from this same distribution (Fig 5B). However, it is possible that the effects we have seen here would be exaggerated if the non-access group were also at a higher risk of being bitten (Fig 5C). There are numerous reasons for and quantifications of the heterogeneities in bite risk for lymphatic filariasis, which have been compared within [36] and between locations [37]. It is therefore feasible that for the



Fig 5. Distribution of mosquito vectors across three theoretical scenarios. A. When the whole population has access to treatment and are included in surveillance, with participation controlled by adherence. Vectors are also distributed evenly across the population, so everyone has the same risk of being bitten/ infected. B. The population is separated based on who has access to treatment and is included in surveillance, but those excluded are not at any higher risk of contracting disease. C. Those excluded are also more likely to be bitten/ infected therefore harbouring more disease than in the group with access to treatment, who are less likely to be bitten. Human outlines attributed to Microsoft.

https://doi.org/10.1371/journal.pntd.0011582.g005

same reason treatment is not provided, an individual may be at higher risk of being bitten for example due to employment, home location, poor bed net usage or lack of screened windows. Such a correlation between adherence and bite risk has been shown to significantly impact the probability of reaching EPHP for lymphatic filariasis [23]. Furthermore, the correlation between treatment access and lack of surveillance may be transient depending on the time delay between the two events.

We also make the assumption that all human hosts are homogeneously mixing with the vectors regardless of treatment access and inclusion in surveillance. The assumption of homogeneous mixing is common, however there is evidence that accounting for non-homogenous mixing can explain maintained infection prevalence [38]. Whilst it is likely that a non-homogenous mixing assumption would better describe the behaviour within the population, the point still stands, that increasing disparities in treatment access could negatively impact treatment programme success. We also note, that as these are generic, stylised simulations, the population age structure may be different to that of a location where lymphatic filariasis may be more common, where often the population is skewed towards a greater number of children and fewer adults. This will impact observations made in the six- and seven-year-old group, where a smaller group is likely more prone to random extinctions and resurgences that do not reflect the dynamics of the wider population. Finally, here we used infection prevalence as a metric for long term intervention success. It is possible that infection intensity will have reduced more significantly. However, there is ongoing debate around the relationship between morbidity from infection and infection intensity quantified by diagnostics that lack sensitivity or are hard to use. Given that the public health targets focus on prevalence we have done so here also but suggest further work can be done to explore insights from infection intensity.

To conclude, we show that exclusion to treatment and from surveillance impacts the probability of reaching public health targets, but in the small proportion of populations that do reach the threshold, it does not impact maintenance. The TAS is an insufficient tool for quantifying population-level dynamics, and finally, it is not possible to identify the presence of exclusion, when observations only occur in the same group being treated. Whilst we did not investigate this here, it may be that indirect measures such as xenomonitoring may provide indirect signals. Adapting models such as this one to inform on this is a valuable next step towards identifying and quantifying groups with limited access to treatment, to better inform intervention programmes and end-game management.

Supporting information

S1 Supplementary material. Table A. Parameter values used for the simulations. Fig A. Distribution of prevalences across the 14,145 simulated populations. Fig B. The relationship between the vector-to-host ratios, k (bite aggregation parameter) and prevalence. Increasing vector-to-host ratio is generally accompanied by a more random distribution of bites–i.e., a larger value of k as the distribution of bites becomes more randomly (Poisson) distributed in the population. Table B. The number of populations used in Fig 4 for each combination of observation group and exclusion proportion (i.e., each line in each figure is represented by a row in the below table). These are the numbers of populations under each scenario that appeared to have reached EPHP after 5 years of treatment. Fig C. Systematic non-access on the y-axis and treatment adherence correlation on the x-axis. These are figures from previous analyses conducted on high and low transmission populations (high = starting prevalence ~ 23% and low ~5%). The deepening colour relates to a decreasing percentage point difference between observed and true infection prevalence in the two settings. (DOCX)

Author Contributions

Conceptualization: Jessica Clark, Katherine Gass, Alison Krentel, T. Déirdre Hollingsworth.

Formal analysis: Jessica Clark.

Funding acquisition: T. Déirdre Hollingsworth.

Investigation: Jessica Clark, T. Déirdre Hollingsworth.

Methodology: Jessica Clark, Emma L. Davis, Joaquin M. Prada, T. Déirdre Hollingsworth.

Project administration: T. Déirdre Hollingsworth.

Resources: T. Déirdre Hollingsworth.

Software: Jessica Clark, Emma L. Davis, Joaquin M. Prada, T. Déirdre Hollingsworth.

Supervision: T. Déirdre Hollingsworth.

Validation: Jessica Clark, Joaquin M. Prada, T. Déirdre Hollingsworth.

Visualization: Jessica Clark.

- Writing original draft: Jessica Clark, Emma L. Davis, Joaquin M. Prada, Katherine Gass, Alison Krentel, T. Déirdre Hollingsworth.
- Writing review & editing: Jessica Clark, Emma L. Davis, Joaquin M. Prada, Katherine Gass, Alison Krentel, T. Déirdre Hollingsworth.

References

- 1. World Health Organization. Ending the neglect to attain the sustainable development goals. Geneva: World Health Organization; 2020.
- Bottomley C, Isham V, Vivas-Martinez S, Kuesel AC, Attah SK, Opoku NO, et al. Modelling Neglected Tropical Diseases diagnostics: the sensitivity of skin snips for Onchocerca volvulus in near elimination and surveillance settings. Parasites & Vectors. 2016; 9(1):343. https://doi.org/10.1186/s13071-016-1605-3 PMID: 27301567
- 3. COR-NTD, editor Addressing the issue of 'systematic non-compliance' in mass drug administration for lymphatic filariasis. Research Links; 2021; zoom: COR-NTD.
- 4. Adriko M, Faust C, Carruthers L, Moses A, Tukahebwa E, Lamberton P. Low Praziquantel Treatment Coverage for Schistosoma mansoni in Mayuge District, Uganda, Due to the Absence of Treatment

Opportunities, Rather Than Systematic Non-Compliance. Tropical Medicine and Infectious Disease. 2018; 3(4). https://doi.org/10.3390/tropicalmed3040111 PMID: 30297642

- Chami GF, Kabatereine NB, Tukahebwa EM. The division of labour between community medicine distributors influences the reach of mass drug administration: A cross-sectional study in rural Uganda. PLoS Neglected Tropical Diseases. 2019; 13(9):e0007685. <u>https://doi.org/10.1371/journal.pntd</u>. 0007685 PMID: 31483784
- 6. Rajeev M, Guis H, Edosoa GT, Hanitriniaina C, Randrianarijaona A, Mangahasimbola RT, et al. How geographic access to care shapes disease burden: The current impact of post-exposure prophylaxis and potential for expanded access to prevent human rabies deaths in Madagascar. PLoS Neglected Tropical Diseases. 2021; 15(4):e0008821. https://doi.org/10.1371/journal.pntd.0008821 PMID: 33901194
- Hamill LC, Haslam D, Abrahamsson S, Hill B, Dixon R, Burgess H, et al. People are neglected, not diseases: the relationship between disability and neglected tropical diseases. Transactions of the Royal Society of Tropical Medicine and Hygiene. 2019; 113(12):829–34. <u>https://doi.org/10.1093/trstmh/trz036</u> PMID: 31111941
- Ayabina D, Kura K, Toor J, Graham M, Anderson RM, Hollingsworth TD. Maintaining low prevalence of Schistosoma mansoni: modelling the effect of less frequent treatment. Clinical Infectious Diseases. 2021. https://doi.org/10.1093/cid/ciab246 PMID: 33909064
- Toor J, Truscott JE, Werkman M, Turner HC, Phillips AE, King CH, et al. Determining post-treatment surveillance criteria for predicting the elimination of Schistosoma mansoni transmission. Parasites & Vectors. 2019; 12(1):437. https://doi.org/10.1186/s13071-019-3611-8 PMID: 31522690
- Anderson RM, May RM. Population Dynamics of Human Helminth Infections: Control by Chemotherapy. Nature. 1982; 297:557–63 https://doi.org/10.1038/297557a0 PMID: 7088139
- Babu BV, Babu GR. Coverage of, and compliance with, mass drug administration under the programme to eliminate lymphatic filariasis in India: a systematic review. Transactions of the Royal Society of Tropical Medicine and Hygiene. 2014; 108(9):538–49. <u>https://doi.org/10.1093/trstmh/tru057</u> PMID: 24728444
- Shuford KV, Turner HC, Anderson RM. Compliance with anthelmintic treatment in the neglected tropical diseases control programmes: a systematic review. Parasites & Vectors. 2016; 9:29. <u>https://doi.org/10. 1186/s13071-016-1311-1</u> PMID: 26813098
- Farrell SH, Truscott JE, Anderson RM. The importance of patient compliance in repeated rounds of mass drug administration (MDA) for the elimination of intestinal helminth transmission. Parasites & Vectors. 2017; 10(1):291. https://doi.org/10.1186/s13071-017-2206-5 PMID: 28606164
- Dyson L, Stolk WA, Farrell SH, Hollingsworth TD. Measuring and modelling the effects of systematic non-adherence to mass drug administration. Epidemics. 2017; 18:56–66. <u>https://doi.org/10.1016/j.epidem.2017.02.002</u> PMID: 28279457
- Cabral S, Bonfim C, Oliveira R, Oliveira P, Guimaraes T, Brandao E, et al. Knowledge, attitudes and perceptions regarding lymphatic filariasis: study on systematic noncompliance with mass drug administration. Revista do Instatuto de Medicina Tropical de Sao Paulo. 2017; 59:e23. https://doi.org/10.1590/ S1678-9946201759023 PMID: 28443941
- Coulibaly JT, Ouattara M, Barda B, Utzinger J, N'Goran EK, Keiser J. A Rapid Appraisal of Factors Influencing Praziquantel Treatment Compliance in Two Communities Endemic for Schistosomiasis in Cote d'Ivoire. Tropical Medicine and Infectious Disease. 2018; 3(2). <u>https://doi.org/10.3390/</u> tropicalmed3020069 PMID: 30274465
- Manyeh AK, Ibisomi L, Ramaswamy R, Baiden F, Chirwa T. Exploring factors affecting quality implementation of lymphatic filariasis mass drug administration in Bole and Central Gonja Districts in Northern Ghana. PLoS Neglected Tropical Diseases. 2020; 14(8):e0007009. <u>https://doi.org/10.1371/journal.pntd.0007009</u> PMID: 32804967
- Colt S, Jarilla B, Baltazar P, Tallo V, Acosta LP, Wu HW, et al. Effect of maternal praziquantel treatment for Schistosoma japonicum infection on the offspring susceptibility and immunologic response to infection at age six, a cohort study. PLoS Neglected Tropical Diseases. 2021; 15(4):e0009328. <u>https://doi.org/10.1371/journal.pntd.0009328</u> PMID: 33861768
- Ndibazza J, Muhangi L, Akishule D, Kiggundu M, Ameke C, Oweka J, et al. Effects of Deworming during Pregnancy on Maternal and Perinatal Outcomes in Entebbe, Uganda: A Randomized Controlled Trial. Clinical Infectious Diseases. 2010; 50(4):531–40. https://doi.org/10.1086/649924 PMID: 20067426
- Niles RA, Thickstun CR, Cox H, Dilliott D, Burgert-Brucker CR, Harding-Esch EM, et al. Assessing factors influencing communities' acceptability of mass drug administration for the elimination of lymphatic filariasis in Guyana. PLoS Neglected Tropical Diseases. 2021; 15(9):e0009596. https://doi.org/10.1371/journal.pntd.0009596 PMID: 34543269

- Rilkoff H, Tukahebwa EM, Fleming FM, Leslie J, Cole DC. Exploring gender dimensions of treatment programmes for neglected tropical diseases in Uganda. PLoS Neglected Tropical Diseases. 2013; 7(7): e2312. https://doi.org/10.1371/journal.pntd.0002312 PMID: 23875047
- Irvine MA, Hollingsworth TD. Making Transmission Models Accessible to End-Users: The Example of TRANSFIL. PLoS Neglected Tropical Diseases. 2017; 11(2):e0005206. <u>https://doi.org/10.1371/journal.pntd.0005206 PMID: 28151997</u>
- Irvine MA, Reimer LJ, Njenga SM, Gunawardena S, Kelly-Hope L, Bockarie M, et al. Modelling strategies to break transmission of lymphatic filariasis—aggregation, adherence and vector competence greatly alter elimination. Parasites & Vectors. 2015; 8:547. https://doi.org/10.1186/s13071-015-1152-3 PMID: 26489753
- World Health Organization. Lymphatic filariasis: monitoring and epidemiological assessment of mass drug administration. Geneva; 2011.
- Burgert-Brucker CR, Zoerhoff KL, Headland M, Shoemaker EA, Stelmach R, Karim MJ, et al. Risk factors associated with failing pre-transmission assessment surveys (pre-TAS) in lymphatic filariasis elimination programs: Results of a multi-country analysis. PLoS Neglected Tropical Diseases. 2020; 14(6): e0008301. https://doi.org/10.1371/journal.pntd.0008301 PMID: 32479495
- World Health Organization. Strengthening the assessment of Lymphatic filariasis transmission and documenting the achievment of elimination. In: indicators NTDSaTAGsmaes-god-s, editor. France: World Health Organization; 2016.
- Fimbo AM, Minzi OMS, Mmbando BP, Barry A, Nkayamba AF, Mwamwitwa KW, et al. Prevalence and Correlates of Lymphatic Filariasis Infection and Its Morbidity Following Mass Ivermectin and Albendazole Administration in Mkinga District, North-Eastern Tanzania. Journal of Clinical Medicine. 2020; 9(5). https://doi.org/10.3390/jcm9051550 PMID: 32455556
- Irvine MA, Stolk WA, Smith ME, Subramanian S, Singh BK, Weil GJ, et al. Effectiveness of a triple-drug regimen for global elimination of lymphatic filariasis: a modelling study. The Lancet Infectious Diseases. 2017; 17(4):451–8. https://doi.org/10.1016/S1473-3099(16)30467-4 PMID: 28012943
- Stolk WA, de Vlas SJ, Borsboom GJ, Habbema JD. LYMFASIM, a simulation model for predicting the impact of lymphatic filariasis control: quantification for African villages. Parasitology. 2008; 135 (13):1583–98. https://doi.org/10.1017/S0031182008000437 PMID: 19006602
- Coffeng LE, Stolk WA, Hoerauf A, Habbema D, Bakker R, Hopkins AD, et al. Elimination of African onchocerciasis: modeling the impact of increasing the frequency of ivermectin mass treatment. PLoS One. 2014; 9(12):e115886. https://doi.org/10.1371/journal.pone.0115886 PMID: 25545677
- Rock KS, Torr SJ, Lumbala C, Keeling MJ. Quantitative evaluation of the strategy to eliminate human African trypanosomiasis in the Democratic Republic of Congo. Parasites & Vectors. 2015; 8:532. https://doi.org/10.1186/s13071-015-1131-8 PMID: 26490248
- Coffeng LE, Bakker R, Montresor A, de Vlas SJ. Feasibility of controlling hookworm infection through preventive chemotherapy: a simulation study using the individual-based WORMSIM modelling framework. Parasites & Vectors. 2015; 8:541. https://doi.org/10.1186/s13071-015-1151-4 PMID: 26489659
- Liyew EF, Chernet M, Belay H, Maddren R, Landeryou T, Kalahasti S, et al. Coverage evaluation surveys following soil-transmitted helminthiasis and schistosomiasis mass drug administration in Wolaita Zone of Ethiopia-The Geshiyaro project. PLoS One. 2021; 16(12):e0260722. https://doi.org/10.1371/journal.pone.0260722 PMID: 34932602
- Krentel A, Flscher PU, Weil GJ. A Review of Factors That Influence Individual Compliance with Mass Drug Administration for Elimination of Lymphatic Filariasis. PLoS Neglected Tropical Diseases. 2013; 7 (11). https://doi.org/10.1371/journal.pntd.0002447 PMID: 24278486
- Banda GT, Deribe K, Davey G. How can we better integrate the prevention, treatment, control and elimination of neglected tropical diseases with other health interventions? A systematic review. BMJ Glob Health. 2021; 6(10). https://doi.org/10.1136/bmjgh-2021-006968 PMID: 34663634
- Davis EL, Reimer LJ, Pellis L, Hollingsworth TD. Evaluating the Evidence for Lymphatic Filariasis Elimination. Trends in Parasitology. 2019; 35(11):860–9. https://doi.org/10.1016/j.pt.2019.08.003 PMID: 31506245
- Irvine MA, Kazura JW, Hollingsworth TD, Reimer LJ. Understanding heterogeneities in mosquito-bite exposure and infection distributions for the elimination of lymphatic filariasis. Proceedings of the Royal Society B: Biological Sciences. 2018; 285(1871). <u>https://doi.org/10.1098/rspb.2017.2253</u> PMID: 29386362
- de Vos AS, Stolk WA, de Vlas SJ, Coffeng LE. The effect of assortative mixing on stability of low helminth transmission levels and on the impact of mass drug administration: Model explorations for onchocerciasis. PLoS Neglected Tropical Diseases. 2018; 12(10):e0006624. <u>https://doi.org/10.1371/journal.pntd.0006624</u> PMID: 30296264